Author's response to reviews

Title: Heated indoor swimming pools, infants, and the pathogenesis of adolescent idiopathic scoliosis: a neurogenic hypothesis

Authors:

Marianne E McMaster Mrs. (m.mcmaster1@btinternet.com)

Version: 4 Date: 23 September 2011

Author's response to reviews:

Formatting changes made 23.9.11:
-------------------------------------------

Major revisions

Title: please ensure that the title is identical in the manuscript and the submission system. COMPLETE

Abstract: Please ensure the abstract is identical in the manuscript and the submission system. COMPLETE

Figures: Please provide a figure title. This should be concise and be the first sentence of the figure legend. Please include this in a figure legends section immediately after the references. Please also refer to this figure as ‘Figure 1’ in the text. COMPLETE

References: please provide a functioning link to website for reference 38. COMPLETE (the link has been removed online).

Consent: If possible, please include a consent statement for Figure 1 including the following information 'Written informed consent was obtained for accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.' COMPLETE. "This illustration was taken from our teaching archive. The individual cannot be identified and cannot be recognized from the photograph"

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above). COMPLETE

Key words: Please remove the key words list from the abstract: COMPLETE

References: If now possible, please provide the full citation details for reference 50. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references. COMPLETE